Author's response to reviews

Title: External metallic circle in hepaticojejunostomy

Authors:

- Erdal Gocmen (mesuttez@yahoo.com)
- Mehmet Keskek (mesuttez@yahoo.com)
- Mesut Tez (mesuttez@yahoo.com)
- Sebat Karamursel (mesuttez@yahoo.com)
- Mahmut Koc (mesuttez@yahoo.com)
- Mehmet Kilic (mesuttez@yahoo.com)

Version: 7 Date: 30 September 2004

Author's response to reviews:

The figure was cropped. The e-mail addresses for all authors were added on title page and key words were extracted. The blank page 11 was removed.